

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Olbrich, Roland; Quaas, Martin F.; Baumgärtner, Stefan

Working Paper Characterizing commercial cattle farms in Namibia: Risk, management and sustainability

Working Paper Series in Economics, No. 248

Provided in Cooperation with: Institute of Economics, Leuphana Universität Lüneburg

Suggested Citation: Olbrich, Roland; Quaas, Martin F.; Baumgärtner, Stefan (2012) : Characterizing commercial cattle farms in Namibia: Risk, management and sustainability, Working Paper Series in Economics, No. 248, Leuphana Universität Lüneburg, Institut für Volkswirtschaftslehre, Lüneburg

This Version is available at: https://hdl.handle.net/10419/73177

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

PAPE **NGRKING**

Characterizing commercial cattle farms in Namibia: risk, management and sustainability

by Roland Olbrich, Martin F. Quaas and Stefan Baumgärtner

University of Lüneburg Working Paper Series in Economics

No. 248

August 2012

www.leuphana.de/institute/ivwl/publikationen/working-papers.html

ISSN 1860 - 5508

Characterizing commercial cattle farms in Namibia: risk, management and sustainability

Roland Olbrich^{a*}, Martin F. Quaas^b and Stefan Baumgärtner^a

^a Department of Sustainability Sciences and Department of Economics, Leuphana University of Lüneburg, Germany

^b Department of Economics, University of Kiel, Germany

Abstract

Commercial cattle farming in Namibia, a prime example of livestock farming in semi-arid rangelands, is subject to a variety of risks, predominant among which is precipitation risk. At the same time it suffers from rangeland degradation that is at least partly due to inadequate management. We characterize cattle farms through descriptive statistics and cluster analysis using data that we elicited in August 2008 through mail-in questionnaires and in-field experiments. We find that cattle farms are highly heterogeneous in the majority of individual characteristics. Heterogeneity is also observed when analyzing characteristics jointly through the cluster analysis which suggests classification of farms into three distinct clusters. This classification is predominantly driven by environmental condition and financial risk management, and to a lesser extent by organizational structure of farms or ethnicity. Overall, our study is the first to provide a comprehensive characterization of this system in respect to risk, management and sustainability.

Keywords: cattle farming, semi-arid rangelands, Namibia, empirical survey, perceived risk, management, risk and time preferences, normative views, sustainability

JEL-Classification: Q12, Q15, Q57

^{*} Address for correspondence: Sustainability Economics Group, Leuphana University of Lüneburg, P.O. Box 2440, D-21314 Lüneburg, Germany. Phone: +49.4131.677-2530, fax: +49.4131.677-1381, email: olbrich@uni.leuphana.de.

1. Introduction

Livestock farming is the dominant land use in semi-arid areas. These areas are globally very important, covering 15% of the Earth's surface and housing approximately 900 million people (MEA, 2005: 627). The demanding environmental conditions, chiefly among them low and highly risky precipitation, place serious constraints on rain-fed crop farming. Thus, land is instead used as rangeland for livestock farming, as this offers sufficient flexibility to adapt to environmental conditions. Livestock farming is conducted on 1.2 billion hectares in semi-arid areas which accounts for more than 60% of the land surface used for agricultural purposes (MEA, 2005: 627). However, while livestock farming is intended to deal with the variable environmental conditions it is frequently unsustainable with at least 10-20% of semi-arid areas, including rangelands, being degraded (MEA, 2005: 637, 640). One reason is the use of inadequate management strategies (e.g. Fynn and O'Connor, 2000; de Klerk, 2004; Wiegand, 2010). This is the case not only in communal livestock farming where rangeland is a common pool resource and where it may be rational for farmers to "produce outcomes that are not in anyone's long-term interest" (Ostrom, 1999: 279); it is also the case in commercial livestock farming (de Klerk, 2004) where property-owning farmers exclusively manage rangeland and may do so for decades.

In this paper we empirically characterize one prime example of commercial livestock farming in semi-arid rangelands, which is commercial cattle farming in Namibia. Like other semi-arid rangelands, those in Namibia are also subject to high precipitation risk (Sweet, 1998) and suffer from degradation which comes in the form of bush encroachment (de Klerk, 2004). At the same time, commercial cattle farming is economically important, contributing the largest share to Namibia's agricultural output (MAWF, 2009: 7). Some aspects of commercial cattle farming have been studied extensively, such as bush encroachment (e.g. de Klerk, 2004), but many other aspects relating to (environmental) risk, management and sustainability have been examined to a much lesser extent (see Olbrich (2012), pp. 17-52, for a review). We take a first step to fill this gap by presenting comprehensive descriptive statistics for farmers and farms on 1) personal and farm features, 2) risk perception, 3) risk management strategies, 4) individual risk and time preferences, and 5) normative views of sustainability. We also conduct a cluster analysis to determine whether cattle farms may be classified into similar groups. Data for this study derives from a survey, consisting of a mail-in questionnaire and infield experiments, that we conducted among 2,119 commercial cattle farmers in Namibia in August/September 2008 (Olbrich et al., in press). 398 questionnaires were returned, equaling

a return rate of 19%. We complement this quantitative data by qualitative ones that we collected in personal interviews during the period 2007–2010.

Our results show that cattle farms are highly heterogeneous in the majority of individual characteristics as indicated by high standard deviations. A cluster analysis reveals that cattle farms are also heterogeneous when considering characteristics jointly as exemplified by the identification of three clusters. Classification is predominantly driven by environmental conditions and financial risk management and to a lesser extent by organizational structure of the farms or ethnicity: the most distinct of the three identified clusters is best characterized by high grazing capacity, low perceived rainfall risk and low self-reported financial risk management; of the remaining two clusters, one is best characterized by a high proportion of multiple owners, the other by a high proportion of Afrikaans farmers. Risk and time preferences and normative views of sustainability play only a marginal role for classification. Income does not drive classification and neither does weekend farming, a commonly used classification characteristic in Namibia.

The paper is organized as follows: Section 2 gives a brief description of the background of commercial cattle farming in Namibia. Section 3 describes the data collection. Section 4 details the analytical procedure and results for the descriptive statistics, while Section 5 does so for the cluster analysis. Finally, Section 6 discusses and concludes. Figures and tables are relegated to Appendices A to C (for the descriptive statistics) and D (for the cluster analysis).

2. Background on commercial cattle farming in Namibia

Giving a comprehensive background of commercial cattle farming in Namibia is beyond the scope of this paper, but is given by Olbrich (2012, pp. 17–52). The commercial cattle farming area covers approximately 14.5 million hectares (ha) (Mendelsohn, 2006: 42) and is located in the northern half of Namibia (Figure 1), confined to the south and west by areas too dry for farming and in the north and east by communal lands. On average, the commercial farming area receives an annual precipitation of 374 mm, with 95% (352 mm) of precipitation falling during the rainy season from November to April (NMS, unpublished).¹ Precipitation is highly variable from one rainy season to the next: the coefficient of variation of total rainy season

¹ We refer here to the meteorological year, which is commonly defined from July to June in southern Africa (e.g. Unganai, 1996; Burke, 1997). We define the rainy season as the period 01st of November until 30th of April.

precipitation amounts to 0.35 (NMS, unpublished).² The land is partly degraded, with many farms nowadays having a grazing capacity which is worse than historic value of above 0.1 Large Stock Unit per hectare (LSU/ha) (below 10 ha/LSU)³ that was encountered on average across Namibia until the mid 1960s (de Klerk, 2004: 21).

Commercial cattle farming is economically the dominant livestock farming system in Namibia: it contributes by far the largest share of total agricultural output and approximately 1–2% directly to GPD (MAWF, 2009: 7, 9).⁴ An estimated 2,500 commercial cattle farmers⁵ keep an average of 840,000 cattle (MAWF, 2009: 13). Of the 298,961 cattle that are on average marketed each year, roughly half (49%) of the cattle are sold as live cattle (almost exclusively as weaners) whereas the other half (51%) are converted to beef (MAWF, 2009: 14). Almost all weaners are exported as live cattle to feed lots in South Africa (Schutz, 2010). Beef is primarily sold to South Africa (45%), overseas (37%) and other markets (3%) with only a fraction consumed domestically (15%) (MAWF, 2009: 14, 15). Politically, the majority of commercial cattle farmers are organized in the Namibia Agricultural Union (NAU) which is the main interest group of commercial farmers. The Emerging Commercial Farmers' Support Programme is a smaller interest group that specifically represents indigenous commercial farmers which are typically referred to as "emerging commercial farmers".

3. Data collection

3.1 Quantitative data collection

In August 2008, we conducted a survey consisting of a mail-in questionnaire and in-field risk and time experiments among commercial cattle farmers in Namibia. We elicited characteristics in respect to 1) personal and farm features, 2) risks faced by farmers, 3) risk management strategies, 4) individual risk and time preferences, and 5) normative views of sustainability. Within the questionnaire we employed a variety of question formats, including Likert-scales, multiple-choice questions and open questions. In addition, the questionnaire contained risk and time experiments over hypothetical rewards, parallel to in-field risk and

 $^{^{2}}$ The coefficient of variation for total annual precipitation is with 0.35 very similar. As a reference, we note that for countries in central and northern Europe this coefficient of variation is between 0.1 and 0.2 (Chapman, 2010: Map 2).

³ Throughout this paper, we report grazing capacity in the unit LSU/ha which is more intuitive since higher values then denote better grazing capacity. However, we always also report in brackets the more commonly used inverted value in the unit ha/LSU.

⁴ All subsequent figures from MAWF (2009) are calculated as averages over the period 2000–2007.

⁵ No census data is available that gives the exact number of commercial cattle farmers. The estimate comes from experts of the Namibia Agricultural Union and the Meat Board of Namibia.

time experiments with monetary rewards. A detailed description of the survey can be found in Olbrich et al. (in press) and a copy of the questionnaire is included in Olbrich et al. (2009).

We sent out questionnaires to a group of 2,119 farmers which consisted of members of the NAU and of farmers that deliver cattle to MeatCo, the largest slaughterhouse in Namibia. This group essentially is the whole population of commercial cattle farmers in Namibia (Olbrich et al., in press). We mailed out a first batch of questionnaires in the period $19^{th} - 21^{st}$ of August 2008, and a second batch as a follow up on the 15^{th} of September 2008. In addition, we randomly selected 39 NAU members for participation in in-field risk and time experiments. We visited the majority of participants (79.4%) on their respective farms, and the remaining ones at pubic locations in major cities.

398 questionnaires were returned, equaling a return rate of 19%. In the returned questionnaires, the response rate for non-sensitive questions exceeded 95% for most questions, and the response rate was greater than 90% for sensitive questions such as income. An optional question for identification of the farm was answered by 75.1% of survey participants.

3.2 Qualitative data collection

In addition to the quantitative survey, we conducted qualitative interviews with farmers and decision makers in the agricultural, political and financial sector throughout four research visits in March/April 2007, October 2007, July/August 2008 and February/March 2010. During the last research visit we also conducted a series of workshops with farmers and decision makers to discuss preliminary results of this paper. We will not report findings from qualitative interviews and workshops as such but draw on these findings for the interpretation of some results of the quantitative survey.

4. Descriptive statistics

In a first step, we analyze data through the use of descriptive statistics. Results are presented in the form of figures, summary and frequency tables. Due to the large number of figure and tables we have relegated these to the Appendices: Appendix A contains figures, Appendix B a table of summary statistics and Appendix C frequency tables. In this section, we will explicitly refer to figures and frequency tables, but not to summary statistics (Table 1).

4.1 Personal and farm features

Personal features

We find that the majority of principal decision makers on farms are male (94.7%) (Table 2). Household size is on average 3.7 members with most households (38.2%) consisting of two members (Table 3). A small percentage of households (1.5%) report to consist of eleven or more members, with the largest household indicated as having 46 members. However, we deem it likely that very large figures for household size, i.e. 20 members or higher, are artefacts that arose from a misunderstanding of the question.

Farmers are very heterogeneous in respect to age and the distribution is centred within the advanced age: mean age of farmers is 55.4 years with a standard deviation of 11.9 years. The proportion of farmers that are 35 years or younger amounts to only 4.3%, while the proportion of those older than the typical Namibian retirement age of 65 years is at 20.3% relatively high (Figure 2, Table 4). One possible explanation that emerged in our qualitative data collection is that farmers are accustomed to work until old age which in turn entails that their offspring take over businesses typically at an age well beyond 30 years.

The age distribution is also reflected in the distribution of both farm experience and expected duration until retirement. Average farm experience is 25.4 years with a standard deviation of 14.3 years. At the tails of the distribution we find that only 16.5% of farmers have experience of 10 years or less whereas an almost equally sized proportion (13.2%) has experience of 41 years or more (Table 5). Farmers acquired the majority of their farm experience on their own farms: on average, farmers operated their farm for 21.1 years. Regarding retirement, we find that a third of farmers (31.9%) plan to retire within the next 10 years and almost another third (29.6%) plan to do so in the next 11 to 20 years (Table 6). 20.7% of farmers did not specify a precise duration until retirement but stated that they will continue until they either die or their children take over the business.⁶

Expected reasons for retirement (an open question) are predominantly age-related ones (82.5%) such as frailty or death (Table 7). In contrast, political (9.7%), economic (8.3%) or environmental (2.4%) conditions as well as crime (1.9%) are of only minor importance. Correspondingly, the majority of farmers expect a regular transition of the farm to the next

⁶ Mean age of these 20.7% of farmers is 54.4 years. Thus, we consider it likely that some of these farmers will likewise quit farming within the next 20 years. Thus, the fraction of farmers that expect to quit within this timeframe is probably even higher than the 61.5% who explicitly indicated this.

owner (another open question) which is expected to be a family member (68.2%), an unrelated person who purchases the farm (16.6%) or a new manager (3.5%) (Table 8). Only a minority expects to be expropriated (2.7%), and an even smaller fraction expects the business to be dissolved (2.2%). The low percentage of farmers expecting expropriation of their own farm is at odds with farmers' high rating of this risk in general (see Section 4.2 below) as well as with findings from our qualitative data collection where the possibility of expropriation commands considerable attention among farmers. It thus appears as if farmers feel this risk but do not expect to be personally affected.

Farmers are almost exclusively of Afrikaans (46.1%) or German (45.3%) descent, with indigenous farmers accounting for only 5.3% (Table 9). This distribution of ethnicity reflects that commercial cattle farming was the domain of white famers until Namibia's independence in 1991, and moreover that settlement of commercial farmers occurred historically during German colonial rule and South African administration (Mendelsohn, 2006). Closely connected to the distribution of ethnicity is the classification of farmers as being emerging commercial farmers (13.8%) or not (86.2%) (Table 10). However, we acknowledge a possible bias in both ethnicity and classification as emerging commercial farmer due to our sample design: the main channel for promoting the survey was the NAU in which indigenous farmers are underrepresented. This in turn implies that indigenous farmers (and thus also emerging commercial farmers) may be underrepresented in our sample (see Olbrich et al. (2009) for a detailed discussion).⁷ Finally, 80.9% of farmers are NAU members.

Education is of high importance among farmers (Table 11). In a sector where the next generation of farmers may learn the essential farm management skills by growing up on a farm, we find that only 28.4% of farmers did not receive any post-secondary education. 14.7% have learned a trade and 57% have attended college or university. Among the fields of post-secondary education, agriculture ranks highest (34.8%), followed by engineering (20.3%) and business related studies (12.3%) (Table 12). Education (4.3%) and health (3.6%) are fields of minor importance. Partly responsible for the high level of education may be that 19.9% of farmers operate only on the weekend, i.e. consider farming a secondary occupation

⁷ Based on the definition of emerging commercial farmers, the fraction of emerging commercial farmers (13.8%) should not exceed the fraction of indigenous famers (5.3%). This again suggests an artifact where part of the farmers may have misunderstood either the question referring to ethnicity or, more likely, the question referring to classification as emerging commercial farmers.

or hobby, while earning their livelihood primarily in a field that requires post-secondary education (Table 13).⁸

Farm features

Cattle farms in our sample are well distributed throughout the area considered to be the main commercial cattle farming area (Figure 1) (Mendelsohn, 2006: 43). Most farms are located in the regions Otjozondjupa (35.4%), Omaheke (22.3%) and Khomas (20.5%), as well as in Kunene (9.9%) (Table 14). Only few farms are in Erongo (5.6%), Hardap (2.8%) and Karas (0.8%), which are regions where precipitation conditions are very low and variable (Mendelsohn, 2006). The region Oshikoto (2.8% of farms) is largely communal farmland.

Farmer's assessments of the previous five rainy seasons show temporal variability across seasons. Average ratings on a six-item Likert-scale are above the mean of 3.5 for all seasons except for 2006/07 which was rated at only 2.9. The 2007/08 rainy season, which directly preceded the survey data collection, was rated with 4.7 as very good. Precipitation data which we elicited for the individual farms for the seasons 2006/07 and 2007/08 confirmed the above assessment: the subpar-rated 2006/07 season had an average rainfall of 270 mm which is well below the long-term average of 352 mm for the commercial cattle farming region (NMS, unpublished). In contrast, the highly rated 2007/08 season had on average 439 mm of rainfall. All in all, the temporal pattern in rainy season assessments and on-farm precipitation data conform to the pattern found in the precipitation data set collected by the Namibia Meteorological Service (NMS, unpublished). In addition to temporal variability, seasons were also spatially variable: the standard deviations of the season assessments were 1.2 or higher for all seasons, and the standard deviations of on-farm precipitation were 134 mm and 136 mm for the 2006/07 and 2007/08 seasons, respectively.

Most farms are operated by an owner (91.4%), and only a small fraction by tenants (3.8%) or managers (2.8%) (Table 15). The ownership structure of farms is predominantly ownership by a single owner (69.7%) (Table 16). However, a considerable fraction of farms is organized in some form of joined ownership, which includes corporations (21.5%), partnerships, trusts or foundations (5.2%) and cooperatives (2.1%).

Farms are large, with area of *owned* farmland averaging 7,178 ha. Even the area of *owned* farmland is not sufficient for a substantial fraction of farmers: extra land is rented or used free

⁸ Indeed, the distribution of education is shifted towards higher education among the so called "weekend farmers" as compared to fulltime farmers.

of charge by a third of land-owning farmers $(32.5\%)^9$, resulting in *operated* farmland (i.e. *owned* farmland plus net *rented / used free of charge* farmland) averaging 8,401 ha.¹⁰ Finally, not all of the operated farmland is used as *rangeland* for cattle farming but on average only 7,949 ha with the remaining 452 ha set aside for other farming purposes.¹¹ The large average size of *rangeland* reflects the extensive nature of cattle farming which requires large areas due to the demanding environmental conditions. At the same time, there is considerable heterogeneity in *rangeland* area with a standard deviation of 5,512 ha (Figure 3, Table 25); the same is true for *owned, rented / used free of charge* and *operated* farmland (Table 17 to Table 24). Small farms with *rangeland* area of 2,500 ha or less are relatively uncommon (7.7%) whereas farms above 7,500 ha constitute more than two fifths (42.9%) of all farms. This large heterogeneity may reflect a variety of causes such as spatial heterogeneity in environmental conditions or heterogeneity in individual farm management practices. In terms of internal organization, the *rangeland* of a farm is on average divided into 31.5 camps, which is the basic grazing unit on which cattle are kept for a period of days to weeks.

Not only the quantity of rangeland is heterogeneous across farms, but also its quality. This is less obvious when analysing overall quality of rangeland which is on average rated to be high (4.3 on a six-item Likert-scale) with an intermediate level of the standard deviation of 1.0. A different picture emerges when analysing bush cover: only one farm is classified in the lowest category of bush cover (i.e. no bushes) and only 3.8% of farms in the highest category (i.e. 81 to 100% of rangeland covered), but all other categories from low to high cover are well represented (Table 26). Almost half of the farms (48.2%) have bush cover that is intermediate or higher (i.e. 41% or more of the farm covered by bushes), but only 12.1% of farmers consider this cover to be optimal (Table 27). The majority (58.9%) instead prefers low or no cover (i.e. 0 to 20%). Indeed, 62% of farmers consider the bush cover on their farm to be too high (Table 28). A high bush cover negatively impacts the amount of grazing that is available

⁹ This figure combines the fraction of farmers that have rented land for one year or less and the fraction that have rented land for more than one year.

¹⁰ Curiously, whereas 553,986 ha of land is *rented / used free of charge*, only 74,944 ha are *rented out / given free of charge*. The discrepancy in area of 479,042 ha corresponds in turn to the difference in average owned and operated farmland. We cannot conclusively explain this discrepancy but one possible explanation is that we have undersampled certain segments of the commercial farmer population, specifically the emerging commercial cattle farmers. If indeed emerging commercial farmers rent out part of the "missing" area, this might suggest the existence of an informal market for renting out farmland as land acquired under the land-reform-act (by way of which many emerging commercial farmers acquired their land) must not be rented out.

¹¹ For all farms combined we find that total *owned* farmland amounts to 2,806,503 ha, total *rented / used free of charge* farmland to 553,986 ha, total *operated* farmland to 3,293,045 ha and total *rangeland* to 3,116,073 ha. Total *rangeland* thus constitutes 21.5% of the area of 14,500,000 ha that Mendelsohn (2006), p.42, estimated for commercial cattle farmland in Namibia, a fraction which is remarkably similar to our estimated questionnaire-return rate of 20.8%.

to cattle. Consequently average grazing capacity is 0.080 LSU/ha (14.8 ha/LSU) which is less favourable than the historic 0.01 LSU/ha (10 ha/LSU) that were found on average prior to the mid 1960s (de Klerk, 2004: 21) (Table 29). Like bush cover, grazing capacity is also spatially variable across farms with a standard deviation of 0.040 LSU/ha.

Finally, heterogeneity in farms is also reflected in cattle numbers (Figure 4, Table 30 to Table 32). 458 cattle were on average on each farm in April 2008, i.e. at the end of the rainy season 2007/2008, but the standard deviation of 377 cattle is considerable. Cattle owned by third persons that were kept on the farm constitute with 32 cattle on average roughly 7% of total cattle. Farmers owned with 445 cattle slightly less at the beginning of the rainy season in November 2007 than in April 2008, but the difference in cattle numbers between beginning and end of the season was statistically significant (t-test, p<0.05). Without further data it is, however, impossible to tell if this difference i) is specific to the year of data collection where a drought season was followed by a good rainy season, ii) is a regular pattern or iii) has an altogether different explanation.

In terms of production systems, the sector is dominated by oxen production (47.7%), i.e. the production for sale to a slaughterhouse of 18–24 month old oxen reared from a stock of mother cows (Table 33). Of less importance is weaner production (26.9%), i.e. the production for live sales on auctions of eight months old weaners reared from a stock of mother cows, and speculation production (12.0%), i.e. the production for sale to a slaughterhouse of oxen reared from purchased weaners. Thus, even though environmental conditions are fairly unstable in Namibia, farmers focus on the production system that has the most demanding requirements on environmental condition as feed has to be available throughout the whole production cycle of 27–33 months (which includes the mother cow's pregnancy of 9 months). The likely explanation is that this production system also offers the highest profits (Olbrich, 2012: 17–52).

Farmers predominantly belong to the top Namibian income groups. To conclude this we compare our income data to the latest available national income data that was elicited in 2003/2004 (CBS Namibia, 2006: 38). In order to make this comparison we have to express our data in 2004 prices which we do by adjusting for the consumer price inflation rate of 20% in the period 2004–2008 (CIA, 2011). We then find that those 83.9% of farm households who report an annual net income of more than N\$ 50,000 in 2008 (Table 34) have a corresponding income of N\$ 41,666 in 2004 prices which places them considerably above the median income (i.e. N\$ 29,361). Those 19.6% of farmers that report an income of above N\$ 350,000

have a corresponding income of above N\$ 291,666 in 2004 prices which places them somewhere in the top five percentiles (i.e. those 5% of households that earn more than N\$ 183,227).¹² In terms of income diversification we find that most farmers only partly rely on cattle farming for income, with 70.8% of farmers deriving 80% or less from cattle farming (Table 35). Alternative income sources are on-farm services and products such as tourism (7.8%), small stock (5.9%) and game farming (3.3%) (Table 36), as well as off-farm income which 61.1% of farmers obtain to some degree (Table 37).

4.2 Risks faced by farmers

Of all risks, farmers are most concerned with economic risks. The risk of unfavourable trade agreements on beef export is rated highest (5.3 on a six-item Likert-scale), followed by the risk of unfavourable input prices (5.2), of unfavourable cattle output prices (5.1) and of rising living expenses (5.0). The high ratings of these risks may in part be influenced by events occurring prior to the survey, which included the renegotiation of Namibia's trade agreements with the EU (Meyn, 2007), volatile petroleum prices (DE, 2011) and living expenses (CBS Namibia, 2010: 3) and volatile beef and weaner prices (Meat Board, unpublished), but may also indicate a general wariness of economic risks.

Environmental risks are in part assessed to be high such as the risk of bush fire striking the farm (4.8), the rainfall risk (4.6) or the risk of low groundwater level (4.5), and in part assessed to pose an intermediate threat such as the risk of cattle diseases (3.7) or of cattle losses from predators (3.6). We presume that the latter two risks are rated relatively low since techniques like vaccination or hunting predators exit to mitigate these risks.

Risks pertaining to the political or social situation are likewise assessed to be high. The risk of changing labour market conditions (4.8) might hereby reflect a concern with further restriction on the labour market that make employing and laying-off farm workers more difficult after similar laws have already been passed in recent years. The risk of expropriation is also rated to be high (4.8), reflecting the uncertain situation on how the land reform in Namibia will evolve. However, as already discussed above, the high rating of this risk in general is at odds with what famers expect in regards to the future of their own farm (c.f. Table 8). Risk of cattle theft (4.4) is also high and matches information from qualitative interviews as well as regular media reports of such incidents (e.g. Isaacs, 2007).

¹² The top percentile is at N\$ 339,455.

Lastly, failure of machinery or infrastructure (3.3) is rated lowest of all risk. The likely reason is again that techniques exit to specifically mitigate this risk.

Heterogeneity in risk rating across farmers differs considerably between risks. Standard deviations of ratings for environmental risk such as low groundwater level (standard deviation: 1.7), cattle diseases (1.6) or cattle losses from predators (1.6) are relatively heterogeneous while others such as the market price risks (1.0 to 1.1) are much more homogenous. One explanation is that some risks may differ in their exogenous characteristics at the local scale, which fits well to our findings for most environmental risks. Other risk may be fairly homogenous at the national scale, which in turn may be what we are seeing in the high homogeneity of ratings of economic risks.¹³

4.3 Risk management strategies

Farmers predominantly consider on-farm management strategies to be important in the management of risky pasture production. Especially those on-farm strategies where the decision process is in the hand of farmers are rated high, i.e. resting part of the rangeland as a reserve in good rainy seasons (4.7 on a six-item Likert-scale), purchase of supplementary feed (4.7), choice of breed adapted to high variability in grass production (4.5) and choice of cattle production system (4.4). In contrast, the two remaining on-farm strategies, i.e. purchase/lease of rangeland for spatial diversification (3.3) or scale effects (3.3), are rated considerably lower. We cannot conclusively explain these latter two findings.

Financial risk management strategies are of less importance. Checking accounts as a financial buffer (4.7), income from off-farm employment or assets (4.0) and forward contracts (3.9) are all rated relatively high, which is unsurprising since farmers generally have a checking account, derive at least some off-farm income (see Section 4.1) and frequently conclude forward contracts for cattle sales with MeatCo. In contrast, farmers seem to be sceptical towards the remaining financial management strategies: advances on livestock sales (3.1), loans for covering operating losses (3.0) and investment into agricultural derivatives (2.4) are among the lowest rated strategies.

Collective risk management strategies are a mixed bag. Highest rated among these, and indeed among all risk management strategies, are interest groups at a national level such as the NAU (5.0), presumably because they may address a variety of economic, social and political risks. Government support (3.9) and interest groups at a local level (3.8) are rated at

¹³ For completeness, we note that not only the exogenous characteristics but also a variety of factors relating to endogeneity of risks may determine heterogeneity of risk ratings across farmers (Shogren and Crocker, 1999).

intermediate importance. Cooperative ownership of farmland (2.4) is of low importance, which agrees with the result that this organizational form of farms is adapted for only a small fraction of farms (see Section 4.1 above).

In regards to heterogeneity in ratings across farmers, we find considerable heterogeneity for most risk management strategies (standard deviation of 1.6 to 1.8). Our interpretation of this finding is firstly that farmers differ in the general extent in which they employ risk management and secondly that many individual strategies may be substitutes which leaves the individual farmer considerable leeway in the choice of specific strategies.

4.4 Risk and time preferences

Farmers are predominantly risk averse, as indicated by responses in hypothetical risk experiments in the questionnaire as well as in the in-field experiments (Table 38 and Table 39). In a detailed analysis of the risk experiments, Olbrich et al. (2011b) calculate for the average farmer a point estimate for the coefficient of relative risk aversion (CRRA) of 0.78.¹⁴ This estimate is slightly higher than the value of 0.54 reported for a field study of semisubsistence farmers in Ethiopia, India and Uganda by Harrison et al. (2010), but in range with the value of 0.79 provided for the Danish population by Andersen et al. (2006). In an alternative approach for eliciting risk preferences through a self-assessment question on a nine-item Likert-scale ranging from very risk averse (1) to very risk loving (9), farmers indicate on average a value of 5.3. If one assumes that risk neutrality is located in the exact middle of the Likert-Scale, at a value of 5, then the average value of 5.3 would indicate slight risk attraction. This is at odds with above findings from the risk experiments. However, a detailed analysis on the location of risk neutrality on the Likert-Sale is beyond the scope of this publication and we thus cannot conclude in how far average values calculated from both elicitation approaches agree. Finally, in regards to heterogeneity of risk preferences we find intermediate levels of variability in both the hypothetical and in-field experiments, where the majority of farmers are placed in a few closely connected categories (Table 38 and Table 39), and in the self-assessment question, where the standard deviation amounts to 2.1. Such heterogeneity is not unexpected and has been frequently demonstrated in studies of risk preferences (e.g. Andersen et al., 2006).

Farmers are of intermediate impatience. Responses of farmers in the hypothetical as well as in the in-field time preference experiments are centred on discount rates between 10% and 30%

¹⁴ In Olbrich et al. (2011b), as well as in the subsequently cited papers, a positive value of the CRRA indicates risk aversion, a negative value risk attraction and a value of zero risk neutrality.

(Table 40 and Table 41). The distribution of discount rates is shifted towards slightly higher discount rates as compared to findings from other field studies, such as for the Danish population (Harrison et al., 2005c). Calculating point estimates of discount rates is again beyond the scope of this publication. Hence, we do not report these here. Intermediate impatience is also reflected by responses to an alternative self-assessment question on a nine-item Likert-scale ranging from very impatient (1) to very patient (9) with an average value of 5.5. In regards to heterogeneity of time preferences we find the same picture as for risk preferences: intermediate variability may be found in responses to both hypothetical and infield time experiments as well as in the self-assessment question. Heterogeneity of time preferences has likewise been demonstrated in previous studies (e.g. Harrison et al., 2005c).

4.5 Normative views of sustainability

Our conceptualization and operationalization of sustainability is detailed in Baumgärtner and Quaas (2009), and we briefly summarize it here. We *conceptualize* sustainability as a norm, i.e. as an obligation to "pass on a world of undiminished life opportunities to members of future generations" (Howarth, 2007). From this rather abstract norm, individuals derive normative views¹⁵ of sustainability that are heterogeneous across individuals and provide concrete guidance on how to act. We *operationalize* these normative views through ecological-economic viability (Baumgärtner and Quaas, 2009), a criterion for strong sustainability under uncertainty. Normative views of sustainability then pertain to i) what should be sustained ("objects"), ii) how much at least of each of them ("thresholds"), iii) to what extent of risk ("acceptable risk") and iv) for how long ("time horizon").

Prior to the survey we had already identified in our qualitative interviews that grazing capacity (which may be viewed as a proxy for ecosystem condition) and income are the most relevant objects that should be sustained. Consequently, we elicited in the survey

- sustainable grazing capacity (sustainable income) as the threshold at or above which grazing capacity (income) should be sustained,
- acceptable grazing capacity risk (acceptable income risk) as the acceptable risk that grazing capacity (income) falls below the threshold, and
- time horizon over which grazing capacity and income should be sustained at the respective thresholds.

¹⁵ Alternative terms exist for what we labeled "normative views" in the questionnaire. For example, common in the psychological literature is the term "personal norms" which was coined by Schwartz (1973, 1977) and which we use in Olbrich et al. (2011a).

We find that farmers consider on average that grazing capacity should be sustained at or above a threshold of $0.082 \text{ LSU/ha}^{16}$ and annual net income at or above a threshold of N\$ 275,791. Heterogeneity for both sustainable grazing capacity and income is high with standard deviations of 0.045 LSU/ha and N\$ 206,897 and, respectively.

In regards to the time horizon for sustaining grazing capacity and income we find that 8.7% of farmers do not care about the future beyond their own generation, whereas 16.1% of farmers have a very long outlook, i.e. ten generations or more (Table 42). Most farmers (56.1 %) indicated that grazing capacity and income ought to be sustained for the two generations following their own generation, i.e. for the generations of their children and grandchildren, which is the timeframe that most farmers will be expected to experience in their lifetime.

Acceptable grazing capacity risk and acceptable income risk are both centred at an intermediate value of 0.6.¹⁷ Distributions of both probability thresholds are, however, spread out over the whole range of possible values, as exemplified a standard deviation of 0.2 for both characteristics, revealing large heterogeneity across the farmers' population.

In addition to these normative views pertaining to grazing capacity and income, we asked in an open question what should be sustained besides those two objects. Farmers predominantly indicated groundwater level (36.7%) and ecosystem quality (27.6%), reflecting farmers' worries about falling groundwater levels (see Section 4.2) and about degrading ecosystems (Table 43). Furthermore, the sustainability of the social order in an encompassing way appears to be important as farmers also indicated that social conditions (15.4%), the political conditions (14.3%) and infrastructure (10.5%) should be preserved.

Finally, we note that the normative view pertaining to acceptable income risk is uncorrelated to risk preference (Pearson correlation: r=-0.01, p=0.89, N=359) and that the normative view pertaining to time horizon for sustaining grazing capacity and income is uncorrelated to time preference (Pearson correlation: r=-0.04, p=0.51, N=346). We have not elicited preferences over income, grazing capacity or grazing capacity risk and may thus not conduct likewise correlation analyses that involve these preferences and the respective normative views.

¹⁶ In the questionnaire we elicited the characteristic in the unit ha/LSU. We report the inverted value because higher values then denote more demanding norms. For example, it is more demanding to require that grazing capacity is 0.08 LSU/ha instead of only 0.04 LSU/ha.

¹⁷ In the questionnaire we elicited acceptable grazing capacity risk (acceptable income risk) as in how many out of every ten years it is manageable that income (grazing capacity) falls *below* the specified income (grazing capacity) threshold. We make two adjustments in the way we report these characteristics here: firstly, we recalibrate them so that they now specify the still acceptable risk that grazing capacity (income) is *above* the threshold. Secondly, we express values as probabilities where one year out of every ten years is expressed at a probability value of 0.1, two years as a probability value of 0.2, and so forth.

Nonetheless, we conclude that we find no evidence that normative views are dependent on preferences.

5. Cluster analysis

5.1 Statistical methods

In the previous section we characterized commercial cattle farms by describing individual characteristics separately. We now analyze characteristics jointly through a cluster analysis to explore whether we may classify farms into similar groups. In regards to risks and risk management strategies we focus on the dominant precipitation risk and the on-farm and financial risk management strategies. Thus, we will omit all other risks as well as collective risk management strategies.¹⁸

Specifically, we conduct a hierarchical cluster analysis. We use Ward's method for agglomeration over an N x N dissimilarity matrix, where N is the number of observations (Ward, 1963). The matrix contains as elements the Gower dissimilarity measure between observations which is designed to accommodate both continuous and binary characteristics (Gower, 1971). It is defined as

$$D_{ij} = \frac{\sum_{k} w_{ijk} d_{ijk}}{\sum_{k} w_{ijk}}$$

where D_{ij} is the dissimilarity between observation *i* and *j* as the sum of the dissimilarities d_{ijk} between observation *i* and *j* with respect to each characteristic $k = \{1, ..., K\}$ (StataCorp, 2007; Everitt et al., 2011). w_{ijk} is a binary indicator that takes on the value 1 if observations *i* and *j* have non-missing entries for characteristic *k* and is 0 otherwise. We only include observations that have non-missing entries for all *K* characteristics since all D_{ij} are then calculated over the same set of characteristics. Thus, w_{ijk} always takes on the value 1, and the denominator equals *K*.

The specification of d_{ijk} differs between binary and continuous characteristics. For binary characteristics,

¹⁸ Including these other risks and collective management strategies, which are altogether 16 characteristics, would exact the cluster analysis to classify at least partly according to them instead of only according to the more relevant other characteristics.

$$d_{ijk} = \begin{cases} 0 & if \ x_{ik} = x_{jk} \\ 1 & otherwise \end{cases}$$

where x_{ik} and x_{jk} are the values that characteristic k takes on for observations i and j, respectively. For continuous characteristics,

$$d_{ijk} = \frac{\left|x_{ik} - x_{jk}\right|}{\max(x_k) - \min(x_k)},$$

which standardizes the absolute distance between x_{ik} and x_{jk} by the range of values that characteristic *k* takes on over all observations.

We chose the number of clusters by calculating the pseudo F index (Calinski and Harabasz, 1974), where large values indicate a good number of clusters, and the pseudo T squared Je(2)/Je(1) index ("pseudo T squared index") (Duda and Hart, 1973), where low values indicated a good number of clusters, and by subsequently identifying local maxima and minima, respectively. As a robustness check we require that both indices display local optima at the same number of clusters. Subsequent to the cluster analysis, we examine in regards to which characteristics the clusters differ significantly overall and exactly which clusters are responsible for the significant difference. For continuous characteristics, we thereto conduct one-way analyses of variance (ANOVA) followed by pair-wise, Bonferroni-corrected t-tests. For binary characteristics, we conduct Chi-square tests followed by pair-wise, Bonferroni-corrected Chi-square tests. All analyses are performed using the Stata/SE 10.1 statistical software package.

5.2 Data processing

Not all characteristics we describe in Section 4 are suitable for the cluster analysis in the way we measured them, and we have to transform, aggregate and exclude several characteristics.

We transform categorical measured characteristics into binary characteristics where the values '1' denotes the most frequent category.¹⁹ We transform ordinal measured characteristics into interval-scaled discrete characteristics with the artificial unit "index point" where the value '1' denotes the first category, the value '2' the second category and so forth.²⁰ In regards to the time horizon characteristic we set all values greater than ten generations, such as

¹⁹ For example, the categorical measured characteristic 'ownership structure' (Table 16) is transformed in such a way that the value '1' denotes single owners (the most frequent category, indicated by 69.7% of respondents) whereas the value '0' denotes the other structures that involve multiple owners.

 $^{^{20}}$ For example, the ordinal measured characteristic 'actual bush cover' that has six categories is transformed into a discrete characteristic that takes on the values 1, 2, ..., 6.

"forever", to the value ten generations. In regards to risk and time preferences we construct indices for risk and time preference out of the raw responses to the respective hypothetical experiments in the questionnaire: subjects typically prefer in the risk experiments the lottery (which is in our case the cattle auction) when the certain amount is low and in the time experiments the earlier payment when the later payment is likewise low. They switch once the certain amount or the later payment are deemed high enough, and we take this switch point as an index for risk and time preference, respectively. We thus construct a risk preference index as a discrete characteristics with values in $\{1, 2, ..., 7\}$ where low values denote high risk aversion, i.e. those farmers who already switch to the certain amount when it is still low. Conversely, high values denote risk attraction, i.e. those farmers who only switch to the certain amount when it is high. For the time preference index we construct a discrete characteristic with values in $\{1, 2, ..., 6\}$ where low values denotes patience, i.e. those farmers who only already switch to the later amount when it is low. Conversely, high values denote impatience, i.e. those farmers only never if the later amount is high.²¹

We aggregate all characteristics that denote area of land rented and land rented out, regardless of duration of rental period, into a single characteristic that denotes net rented land, i.e. land rented minus land rented out. We also aggregate the characteristics that denote the rainy season assessment of the previous five seasons into one characteristic that denotes the average assessment across the five seasons.

We exclude those measures for risk and time preferences that we elicited in the in-field risk experiments since they are available for only 39 farmers. We also exclude the measure that we elicited as self-assessed preferences through the Likert-scales since we wish to include only one type of preference measures. Furthermore, we exclude all characteristics that were elicited in open questions. Finally, when calculating the Gower dissimilarity measure, highly correlated characteristics may bias results as the impact of these characteristics on the measure is overemphasized with respect to the remaining characteristics (Backhaus et al., 2006: 550). We alleviate this issue by excluding characteristics that display a correlation coefficient of 0.6 or higher to any other characteristic. After all exclusions, we retain a set of 33 characteristics over which we conduct the cluster analysis (Table 45).²² We report the most important omitted characteristics in the table that displays the results alongside the included

²¹ As Olbrich et al. (2011b) detail, we encountered irregularities for some farmers in the risk experiments, which we treated as artifacts and excluded in our further analyses. Similar irregularities were encountered in the time experiments and the respective observations were likewise excluded.

²² After exclusion, the vast majority of characteristics are weakly correlated: out of the 528 unique characteristics pairs, only 11 show a correlation coefficient of 0.4 or larger.

characteristics but will not explicitly discuss them in the text. Due to exclusion of any farm that has a missing value for at least one included characteristic, we remain with 108 farms for the analysis.

5.3 Results

In reporting results, we upfront note that we make three language simplifications for convenience sake: firstly, we talk about "characteristics of clusters" when we, of course, actually refer to characteristics of the farmers or farms included in the respective clusters; secondly, the values we report are cluster-averaged values, but we do not explicitly refer to them as "averaged"; thirdly, when we state that a cluster is "different" we always mean, unless otherwise noted, that the discussed clusters differ significantly from *all* other clusters.

Both the pseudo F- and the pseudo T-index have optima jointly at a number of three and nine clusters (Figure 5, Table 44). At three clusters the pseudo T-index has a global minimum while the pseudo F-index has only a local maximum. Conversely, at nine clusters the pseudo F-index has a global maximum and the pseudo T-index' only a local minimum. Examining both indices thus does not give a unique solution to the optimal number of clusters. Nonetheless, we report the three cluster solution as the nine cluster solutions has two disadvantages: firstly, it is not very insightful as the number of clusters is so large that individual clusters are distinct in only very few characteristics; secondly, under this solution we encounter clusters with fewer than 7 observations, making the validity of the analysis doubtful due to the low number of observations.

Cluster ENFIMA

This cluster is the smallest in that it contains 26 out of the 108 analyzed farms, but is also the most distinct cluster. It differs significantly from each of the two other clusters in 10 out of the 33 analyzed characteristics (Table 45). It is best described by favourable environment and low financial risk management ("ENFIMA"), for the following reasons: in regards to environmental characteristics, it has the highest grazing capacity (0.089 LSU/ha; p<0.05) and the lowest rating of rainfall risk (4.4 on a six-item Likert-scale; p<0.1), but differs in the latter only at the significance level of 10% and only from one other cluster. It also has the most favourable values for average rainy season assessment and actual bush cover, but differences to the other clusters are not significant. Thus, of the aforementioned environmental

characteristics it is mainly grazing capacity and to a lesser extent (perceived) rainfall risk that makes this cluster distinct.

In regards to risk management, it has the lowest ratings of the three clusters for all financial risk management strategies, albeit the differences are significant only for the strategies *advances on livestock sales* (1.4 on a six-item Likert-scale; p<0.05) and *loans for covering operating losses* (1.5; p<0.01). In contrast, it does not have the lowest ratings of the three clusters for all on-farm strategies but only for three of these strategies: for *purchase/lease of rangeland for scale effects* (2.7; p<0.1), albeit at only the 10% significance level; and for *choice of production system* and *choice of breed* (both 3.9; p<0.05), albeit differing in both strategies only from one other cluster. Thus, as differences are more pronounced and consistent for financial risk management strategies we consider them, and not on-farm management strategies, as the distinctive characteristics of this cluster.

Cluster ENFIMA also has the most demanding normative views pertaining to acceptable grazing capacity risk (probability threshold of 0.7; p<0.05), possibly because farmers in this cluster experience low environmental risk and can thus "afford" this more demanding normative view. Other normative views are not significantly different. Finally, it is distinct in two characteristics which are not obviously related to environmental condition and management: it has the lowest number of household members (2.7 members; p<0.1) and it is the most patient (2.6 index points out of 6, p<0.1), albeit it is again significantly distinct in the latter characteristics from only one other cluster. It does not differ in any other personal and farm features or in risk preferences.

Cluster MULTOWN

Based on the distinctive characteristics of cluster ENFIMA, the remaining two clusters are accordingly characterized by relatively unfavourable environmental conditions and relatively high financial risk management. Beyond this distinction, however, they also have their own distinct characteristics. The next largest cluster with 36 farms is significantly distinct in five such characteristics and best characterized by multiple owners ("MULTOWN") as it has the highest proportion of them (41.7% of single owners, corresponding to 58.3% multiple owners; p<0.01). It also has the highest area of net rented land (2,587 ha, p<0.05) and the highest area of rangeland, although the difference to the other clusters is not significant. We may interpret this as a tenuous indication that multiple owners have the means to operate altogether larger farms. This cluster also differs from the other clusters in characteristics that are less obviously associated with multiple ownership: it has the highest rating of the strategy *advances on*

livestock sales (3.6; p<0.05), the lowest rating of the strategy *resting part of the rangeland* (4.1; p<0.05) and the youngest farmers (46.9 years; p<0.01), albeit it is in the latter two characteristics distinct from only one cluster.

Cluster AFRIKAANS

The largest cluster with 46 farms is distinct in four characteristics. It is difficult to describe this cluster as we see no obvious connection between these characteristics; we opt to describe it as Afrikaans farmers ("AFRIKAANS") as it exclusively consists of farmers of this ethnicity (p<0.01). Beyond this distinction, it has an intermediate rating of the strategy *advances on livestock sales* (2.6; p<0.05) and, differing significantly from one other cluster, has the lowest proportion of oxen production (42.3%; p<0.01) and the lowest education level (3.4 index points; p<0.05). Both clusters MULTOWN and AFRIKAANS are not distinct in preferences or normative views.

Altogether, we thus also observe heterogeneity of cattle farms when classifying them, albeit only one cluster of farms is very distinct. Accordingly to the key distinctive characteristics of this cluster, classification is predominantly driven by environmental characteristics and farmer's management, in the form of financial risk management. To a lesser extent, classification is driven by organizational structure or ethnicity, the defining characteristics of the remaining two clusters. Overall, preferences and normative view play only a marginal role for classification.

6. Discussion and conclusion

We characterize farms for commercial cattle farming in Namibia, a prime example of livestock farming in semi-arid rangelands. We find that cattle farms are highly heterogeneous in a wide range of characteristics, as indicated by high standard deviations. When classifying farms in a cluster analysis according to personal and farm features, perceived rainfall risk, self-reported risk management, risk and time preferences and normative views of sustainability, we also find heterogeneity as exemplified by the identification of three separate clusters. Classification is driven predominantly by environmental conditions and financial risk management and to a lesser extent by organizational structure of farms and ethnicity.

It is interesting to note which characteristics are *not* driving farm classification. Firstly, risk and time preferences and normative views of sustainability are only marginally important for

classification. Based on the observed concomitant differences in management one might hypothesize that preferences and normative views, which are key behavioural determinants, are at least for our case study not related to management behaviour. Regarding preferences, this is controversial and we do not expect that such a hypothesis will be uphold under more in-depth scrutiny than can be achieved through a cluster analysis. Regarding normative views, however, we find indeed no evidence that they impact on farm management in an in-depth analysis (Olbrich et al., 2011a). Secondly, weekend farming, a characteristic typically employed by local farmers and decision makers for farm classification, also does not drive our classification. It thus seems that it is of minor importance in comparison to the farms' environment and management. Finally, income does not differ across clusters. This is curious in the light of the observed differences in environmental condition, specifically in grazing capacity, and in financial risk management, as one might expect that differences in income are associated with differences in the latter characteristics. One possible explanation is that increased financial risk management, while stabilizing income, negatively impacts on grazing condition and thus ultimately leads to a degradation of the system (Quaas and Baumgärtner, 2008).

By means of the last observation we note the limitations of the cluster analysis: it cannot be used to make definite statements concerning the causal relationship between single characteristics and thus cannot be a substitute for an in-depth analysis. Thus, we cannot clarify the exact relationship between environmental condition, financial risk management and income without further analysis, as we have for example done in respect to normative views (Olbrich et al., 2011a).

Altogether, this study is the first to provide a comprehensive characterization of Namibian commercial cattle farms in respect to risk, management and sustainability. It provides the basis for more in-depth analyses of the system, in particular by identifying issues that may warrant close attention. It furthers the understanding of the system and may ultimately contribute to the development of policies that promote sustainability of commercial cattle farming in Namibia.

Acknowledgements

Many scientist, experts and farmers contributed with their comments and discussion to the design of this survey, and we wish to express our gratitude to all of them. Special thanks go to Volker and Ursula Dieckhoff, Arne Gressmann, Claus Hager, Jürgen Hoffmann, Harald

Marggraff, Thomas and Heidrun Peltzer, Elsabe Steenkamp, Peter Zensi and Ibo Zimmermann. We also thank our cooperating organizations Agra Co-operative Ltd., Namibia Agricultual Union and Namibian Agronomic Board. Finally, we are grateful to the German Federal Ministry of Education and Research (BMBF) for financial support under grant 01UN0607.

Appendix A: Descriptive statistics, figures



Figure 1: Commercial cattle farms in Namibia. Dashed line denotes what is considered the commercial cattle farming area (Mendelsohn, 2006). Crosses denote position of all 299 farms which were identified in our mail-in questionnaire.



Figure 2: Distribution of age. N = 395.



Figure 3: Distribution of area of rangeland. N = 392.



Figure 4: Distribution of own cattle on farm in April 2008. N = 358.

Appendix B: Descriptive statistics, table of summary statistics

Table 1: Summary statistics for the areas 1) personal and farm features, 2) risks faced by farmers, 3) risk management strategies, 4) individual risk and time preferences and 5) normative views of sustainability. Displayed are mean, median, standard deviation, minimum and maximum for all continuous and Likert-scale measured characteristics. Characteristics are listed in the order in which they are discussed in the text.

Characteristics	Mean	Median	Std. dev.	Min	Max
1) Personal and farm features					
Household size [number of members]	3.7	3.0	3.7	1.0	46.0
Age [years]	55.4	55.0	11.9	27.0	90.0
Farm experience [years]	25.4	24.0	14.3	0.3	84.0
Duration for operating own farm [year]	21.1	20.0	13.3	0.0	69.0
Rainy season assessment					
[1=very poor, 6=very good]					
2003/04	3.6	4.0	1.2	1.0	6.0
2004/05	3.7	4.0	1.2	1.0	6.0
2005/06	5.0	6.0	1.2	1.0	6.0
2006/07	2.9	3.0	1.5	1.0	6.0
2007/08	4.7	5.0	1.2	1.0	6.0
On-farm precipitation [millimetre]					
Oct 2006	34	29	30	0	200
Nov 2006	36	31	30	0	180
Dec 2006	54	43	46	0	188
Jan 2007	62	55	46	0	330
Feb 2007	44	31	47	0	227
Mar 2007	49	37	46	0	415
Apr 2007	26	19	28	0	230
May–Sept 2007	1.2	0	7	0	73
Total rainy season ²³ 2006/07	270	259	134	13	815
Oct 2007	15	11	17	0	140
Nov 2007	15	11	17	0	102
Dec 2007	26	18	28	0	152
Jan 2008	102	90	70	0	374
Feb 2008	165	155	79	0	463
Mar 2008	113	109	53	0	520
Apr 2008	18	10	28	0	200
May–Sept 2008	3	0	7	0	62
Total rainy season ¹⁷ 2007/08	439	429	136	89	980

²³ The rainy season is defined as 1st of November until 30th of April the following year.

Characteristics (continued)	Mean	Median	Std. dev.	Min	Max
Size of farmland [hectare]					
Land owned	7,178	6,000	5,472	0	57,000
Land rented for one year or less	137	0	654	0	5000
Land rented for more than one year	1,151	0	2,731	0	14,000
Land used free of charge	130	0	923	0	10,500
Land rented out for one year or less	66	0	492	0	5017
Land rented out for more than one year	72	0	433	0	4,600
Land entrusted to others	53	0	497	0	8,500
Land operated	8,401	7,000	6,172	0	64,000
Rangeland	7,949	6,765	5,512	0	44,244
Camps [number of camps]	31.5	24	24.3	1	152
Land quality	4.2	4.0	1.0	1.0	()
[1=very poor quality, 6=very good quality]	4.3	4.0	1.0	1.0	6.0
Grazing capacity [Large Stock Unit per hectare]	0.080	0.077	0.040	0.012	0.500
Cattle on farm [number of cattle]					
Own cattle, Nov 2007	445	368	369	0.0	3200
Own cattle, Apr 2008	458	370	377	0.0	3200
Cattle of third person on farm, Apr 2008	32	0	102	0.0	1238
2) Risks faced by farmers					
[1=no risk, 6=very high risk]					
Environmental risks					
Rainfall	4.6	5.0	1.2	1.0	6.0
Low groundwater level	4.5	5.0	1.7	1.0	6.0
Bush fire	4.8	5.0	1.4	1.0	6.0
Cattle diseases	3.7	4.0	1.6	1.0	6.0
Cattle losses from predators	3.6	4.0	1.6	1.0	6.0
Economic risks					
Unfavourable cattle output prices	5.1	5.0	1.1	1.0	6.0
Unfavourable input prices	5.2	6.0	1.0	1.0	6.0
Rising living expenses	5.0	5.0	1.1	1.0	6.0
Unfavourable trade agreements on beef export	5.3	6.0	1.0	1.0	6.0
Political risks					
Changing labour market conditions	4.8	5.0	1.2	1.0	6.0
Expropriation	4.8	5.0	1.3	1.0	6.0
Cattle theft	4.4	5.0	1.4	1.0	6.0
Failure of machinery or farm equipment	3.3	3.0	1.5	1.0	6.0

Characteristics (continued)	Mean	Median	Std. dev.	Min	Max
3) Risk management strategies					
[1=not at all important, 6=very important]					
On-farm management strategies					
Purchase of supplementary feed	4.7	5.0	1.5	1.0	6.0
Choice of cattle production system	4.4	5.0	1.4	1.0	6.0
Choice of breed adapted to high variability in	4.5	5.0	1.2	1.0	()
grass production	4.5	5.0	1.5	1.0	6.0
Resting part of rangeland in good rainy	4.7	5.0	1.5	1.0	6.0
seasons					
Purchase/lease of rangeland for spatial	2.2	2.0	17	1.0	()
diversification	3.3	5.0	1./	1.0	0.0
Purchase/lease of rangeland for scale effects	3.3	3.0	1.7	1.0	6.0
Financial management strategies					
Forwards contracts	3.9	4.0	1.8	1.0	6.0
Advances on livestock sales	3.1	3.0	1.8	1.0	6.0
Checking account as financial buffer	4.7	5.0	1.4	1.0	6.0
Loans for covering operating losses	3.0	3.0	1.7	1.0	6.0
Income from off-farm employment or assets	4.0	4.0	1.7	1.0	6.0
Investment into agricultural derivatives	2.4	2.0	1.6	1.0	6.0
Collective management strategies					
Cooperative ownership of farmland	2.4	2.0	1.7	1.0	6.0
Governmental support	3.9	4.0	1.8	1.0	6.0
Interest groups on a local level	3.8	4.0	1.6	1.0	6.0
Interest groups on a national level	5.0	5.0	1.3	1.0	6.0
4) Individual risk and time preferences					
Self-assessment of risk preferences					
[1=avoid to take risks, 9=willing to take risks]	5.3	6.0	2.1	1.0	9.0
Self-assessment of time preferences					
[1=not willing to wait, 9= willing to wait]	5.5	6.0	2.5	1.0	9.0
5) Normative views of sustainability					
Sustainable annual net income [N\$]	275,791	240,000	206,896	4,000	2,000,000
Sustainable grazing capacity [LSU/ha]	0.082	0.077	0.045	0.013	0.05
Acceptable income risk [probability]	0.6	0.4	0.2	0.0	1.0
Acceptable grazing capacity risk [probability]	0.6	0.4	0.2	0.0	1.0

Appendix C: Descriptive statistics, frequency tables

Each table contains frequencies per category, give as number of observations, column percentages and cumulative percentages. The exceptions are tables for binary characteristics and for those elicited by open questions which do not contain cumulative percentages.

Gender	No.	Col. %
Female	21	5.3
Male	376	94.7
Total	397	100.0

Table 2: Gender.

Table 3: Household size.

Household size	No.	Col. %	Cum. %	
One member	26	6.6	6.6	
Two members	150	38.2	44.8	
Three members	49	12.5	57.3	
Four members	78	19.8	77.1	
Five members	41	10.4	87.5	
Six to ten members	43	10.9	98.5	
Eleven members or more	6	1.5	100.0	
Total	393	100.0		

Table 4: Age.

Age	No.	Col. %	Cum. %
25 to 30 years	4	1.0	1.0
31 to 35 years	13	3.3	4.3
36 to 40 years	32	8.1	12.4
41 to 45 years	31	7.8	20.3
46 to 50 years	60	15.2	35.4
51 to 55 years	60	15.2	50.6
56 to 60 years	67	17.0	67.6
61 to 65 years	48	12.2	79.7
66 to 70 years	37	9.4	89.1
71 to 75 years	26	6.6	95.7
76 to 80 years	8	2.0	97.7
81 to 85 years	5	1.3	99.0
86 to 90 years	4	1.0	100.0
Total	395	100.0	

Farm experience	No.	Col. %	Cum. %
10 years or less	65	16.5	16.5
11 to 20 years	109	27.6	44.1
21 to 30 years	96	24.3	68.4
31 to 40 years	73	18.5	86.8
41 to 50 years	35	8.9	95.7
51 to 60 years	11	2.8	98.5
61 to 70 years	4	1.0	99.5
71 years or more	2	0.5	100.0
Total	395	100.0	

Table 5: Farm experience. Characteristic is calculated as the sum of years spent as owner, manager and in other functions on the farm.

Table 6: Duration until retirement.

Duration until	No.	Col. %	Cum. %
retirement			
10 years or less	122	31.9	31.9
11 to 20 years	113	29.6	61.5
21 to 30 years	34	8.9	70.4
21 to 40 years	15	3.9	74.3
40 years or more	6	1.6	75.9
Until death	73	19.1	95.0
Until children take over	6	1.6	96.6
Do not know	13	3.4	100.0
Total	382	100.0	

Table 7: Expected reasons for retirement. This was an open question in the survey, where more than one answer was possible. 372 farmers gave at least one answer. Reported are the five most frequent answers.

Reasons for retirement	No.	Col. %
Age related (e.g. frailty,	307	82.5
death)	507	02.0
Political conditions	36	9.7
Economic conditions	31	8.3
Environmental conditions	9	2.4
Crime	7	1.9

Fates of farm	No.	Col. %
Continued by family	251	68.2
Sold	61	16.6
New manager	13	3.5
Expropriated	10	2.7
Dissolved	8	2.2

Table 8: Fates of farm at retirement. This was an open question in the survey, where more than one answer was possible. 368 farmers gave at least one answer. Reported are the five most frequent answers.

Table 9: Ethnicities.

Ethnicities	No.	Col. %	Cum. %
Afrikaans	182	46.1	46.1
German	179	45.3	91.4
Other Caucasian	13	3.3	94.7
Indigenous	21	5.3	100.0
Total	395	100.0	

 Table 10: Classification as emerging commercial farmer.

Emerging commercial	No.	Col. %	
farmer			
Yes	53	13.8	
No	331	86.2	
Total	384	100.0	

Table 11: Education levels.

Education levels	No.	Col. %	Cum. %	
No high school	19	48	4.8	
graduation	17	т.0	7.0	
High school graduation	93	23.5	28.4	
Trade/apprenticeship	58	14.7	43.0	
Diploma/Bachelor	172	43.5	86.6	
Master	42	10.6	97.2	
Doctoral	11	2.8	100.0	
Total	395	100.0		

Main fields of study	No.	Col. %	Cum. %	
Agriculture	96	34.8	34.8	
Business	34	12.3	47.1	
Education	12	4.3	51.4	
Engineering	56	20.3	71.7	
Health	10	3.6	75.4	
Other fields	68	24.6	100.0	
Total	276	100.0		

 Table 12: Main fields of study.

Table 13: Weekend farmers. Those farmers are denoted as weekend farmers who do not live on the farm during the week.

Weekend farmer	No.	Col. %	
Yes	79	19.9	
No	317	80.1	
Total	396	100.0	

Table 14:	Regional	locations	of farms.
-----------	----------	-----------	-----------

Regional locations of	No.	Col. %	Cum. %	
farms				
Erongo	22	5.6	5.6	
Hardap	11	2.8	8.4	
Karas	3	0.8	9.2	
Khomas	81	20.5	29.7	
Kunene	39	9.9	39.6	
Omaheke	88	22.3	61.9	
Oshikoto	11	2.8	64.6	
Otjozondjupa	140	35.4	100.0	
Total	395	100.0		

Status of operator	No.	Col. %	Cum. %	
Owner	360	91.4	91.4	
Manager	11	2.8	94.2	
Tenant	15	3.8	98.0	
Other status	8	2.0	100.0	
Total	394	100.0		

Ownership structures	No.	Col. %	Cum. %
Single owner	269	69.7	69.7
Partnership/trust/	20	5.0	74.0
foundation	20	5.2	/4.9
Cooperative	8	2.1	76.9
Corporation	83	21.5	98.4
Other structures	6	1.6	100.0
Total	386	100.0	

 Table 16: Ownership structures.

 Table 17: Owned farmland. Area is measured in hectare (ha).

Owned farmland	No.	Col. %	Cum. %	
0 ha	23	5.9	5.9	
1 to 2,500 ha	23	5.9	11.8	
2,501 to 5,000 ha	90	23.0	34.8	
5,001 to 7,500 ha	115	29.4	64.2	
7,501 to 10,000 ha	72	18.4	82.6	
10,001 to 20,000 ha	62	15.9	98.5	
20,001 or more	6	1.5	100.0	
Total	391	100.0		

Table 18: Farmland rented for one year or less. Area is measured in hectare (ha).

Farmland rented for one	No.	Col. %	Cum. %	
year or less				
0 ha	366	93.6	93.6	
1 to 2,500 ha	17	4.3	98.0	
2,501 to 5,000 ha	8	2.0	100.0	
Total	391	100.0		

Table 19: Farmland rented for more than one year. Area is measured in hectare (ha).

Farmland rented for	No.	Col. %	Cum. %	
more than one year				
0 ha	299	76.5	76.5	
1 to 2,500 ha	31	7.9	84.4	
2,501 to 5,000 ha	33	8.4	92.8	
5,001 to 7,500 ha	9	2.3	95.1	
7,501 to 10,000 ha	4	1.0	96.2	
10,001 to 20,000 ha	15	3.8	100.0	
Total	391	100.0		

Farmland used free of	No.	Col. %	Cum. %	
charge				
0 ha	381	97.4	97.4	
1 to 2,500 ha	2	0.5	98.0	
2,501 to 5,000 ha	4	1.0	99.0	
5,001 to 7,500 ha	3	0.8	99.7	
10,001 to 20,000 ha	1	0.3	100.0	
Total	391	100.0		

Table 20: Farmland used free of charge. Area is measured in hectare (ha).

Table 21: Farmland rented out for one year or less. Area is measured in hectare (ha).

Farmland rented out for	No.	Col. %	Cum. %	
one year or less				
0 ha	380	97.2	97.2	
1 to 2,500 ha	7	1.8	99.0	
2,501 to 5,000 ha	3	0.8	99.7	
5,001 to 7,500 ha	1	0.3	100.0	
Total	391	100.0		

Table 22: Farmland rented out for more than one year. Area is measured in hectare (ha).

Farmland rented out for	No.	Col. %	Cum. %	
more than one year				
0 ha	375	95.9	95.9	
1 to 2,500 ha	13	3.3	99.2	
2,501 to 5,000 ha	3	0.8	100.0	
Total	391	100.0		

Table 23: Farmland entrusted to others free of charge. Area is measured in hectare (ha).

Farmland entrusted to	No.	Col. %	Cum. %	
others free of charge				
0 ha	381	97.4	97.4	
1 to 2,500 ha	8	2.0	99.5	
2,501 to 5,000 ha	1	0.3	99.7	
7,501 to 10,000 ha	1	0.3	100.0	
Total	391	100.0		

Operated farmland	No.	Col. %	Cum. %	
0 ha	1	0.3	0.3	
1 to 2,500 ha	26	6.6	6.9	
2,501 to 5,000 ha	88	22.4	29.3	
5,001 to 7,500 ha	101	25.8	55.1	
7,501 to 10,000 ha	75	19.1	74.2	
10,001 to 20,000 ha	87	22.2	96.4	
20,001 or more	14	3.6	100.0	
Total	392	100.0		

 Table 24: Operated farmland. Area is measured in hectare (ha).

Table 25: Rangeland. Area is measured in hectare (ha).

Rangeland	No.	Col. %	Cum. %	
0 ha	1	0.3	0.3	
1 to 2,500 ha	29	7.4	7.7	
2,501 to 5,000 ha	97	24.7	32.4	
5,001 to 7,500 ha	97	24.7	57.1	
7,501 to 10,000 ha	77	19.6	76.8	
10,001 to 20,000 ha	81	20.7	97.4	
20,001 or more	10	2.6	100.0	
Total	392	100.0		

Table 26: Actual bush cover. Cover measured as percentage of rangeland covered.

Actual bush cover	No.	Col. %	Cum. %	
0%	1	0.3	0.3	
1 to 20%	79	20.3	20.5	
21 to 40%	122	31.3	51.8	
41 to 60%	107	27.4	79.2	
61 to 80%	66	16.9	96.2	
81 to 100%	15	3.8	100.0	
Total	390	100.0		

Optimal bush cover	No.	Col. %	Cum. %	
0%	9	2.4	2.4	
1 to 20%	215	56.6	58.9	
21 to 40%	110	28.9	87.9	
41 to 60%	34	8.9	96.8	
61 to 80%	11	2.9	99.7	
81 to 100%	1	0.3	100.0	
Total	380	100.0		

Table 27: Optimal bush cover. Cover measured as percentage of rangeland covered.

Table 28: Actual versus optimal bush cover. Cover measured as percentage of rangeland covered. Cells report frequencies in percent. Shaded are cells where actual exceeds optimal bush cover.

				Ac	tual bush co	over		
			1 to	21 to	41 to	61 to	81 to	
		0%	20%	40%	60%	80%	100%	Total
	0%	0.3	0.5	0.8	0.5	0.3	0.0	2.4
ver	1 to 20%	0.0	18.0	18.5	13.5	5.3	1.3	56.6
h co	21 to 40%	0.0	1.9	9.8	9.5	6.3	1.3	28.8
snq	41 to 60%	0.0	0.0	1.9	2.9	3.4	0.8	9.0
mal	61 to 80%	0.0	0.0	0.0	1.3	1.6	0.0	2.9
Opti	81 to 100%	0.0	0.0	0.0	0.0	0.0	0.3	0.3
	Total	0.3	20.4	31.0	27.8	16.9	3.7	100.0

Table 29: Grazing capacity. Grazing capacity is measured as hectare per Large Stock Unit (ha/LSU).

Grazing capacity	No.	Col. %	Cum. %	
5 ha/LSU or less	4	1.1	1.1	
6 to 10 ha/LSU	110	29.0	30.1	
11 to 15 ha /LSU	157	41.4	71.5	
16 to 20 ha /LSU	69	18.2	89.7	
21 to 25 ha/LSU	18	4.7	94.5	
25 to 30 ha/LSU	16	4.2	98.7	
30 ha/LSU or more	5	1.3	100.0	
Total	379	100.0		

Own cattle in November	No.	Col. %	Cum. %	
2007				
No cattle	3	0.8	0.8	
100 or less cattle	36	9.6	10.4	
101 to 250 cattle	84	22.4	32.8	
251 to 500 cattle	135	36.0	68.8	
501 to 750 cattle	64	17.1	85.9	
750 or more cattle	53	14.1	100.0	
Total	375	100.0		

Table 30: Own cattle in November 2007.

Table 31: Own cattle in April 2008.

Own cattle in April 2008	No.	Col. %	Cum. %	
No cattle	4	1.1	1.1	
100 or less cattle	30	8.4	9.5	
101 to 250 cattle	75	20.9	30.4	
251 to 500 cattle	122	34.1	64.5	
501 to 750 cattle	78	21.8	86.3	
750 or more cattle	49	13.7	100.0	
Total	358	100.0		

Table 32: Cattle of third persons on farm in April 2008.

Cattle of third persons	No.	Col. %	Cum. %	
on farm in April 2008				
No cattle	277	72.3	72.3	
25 or less cattle	35	9.1	81.5	
26 to 50 cattle	17	4.4	85.9	
51 to 100 cattle	17	4.4	90.3	
101 to 250 cattle	26	6.8	97.1	
251 or more cattle	11	2.9	100.0	
Total	383	100.0		

Production systems	No.	Col. %	Cum. %	
Weaner	87	26.9	26.9	
Speculation	39	12.0	38.9	
Oxen	154	47.7	86.6	
Stud breeding	32	9.8	96.4	
Other systems	12	3.6	100.0	
Total	324	100.0		

Table 33: Production systems.

Table 34: Annual net income. Income is measured as annual net income in Namibian dollar (N\$) in the period 1^{st} March 2007 to 29^{th} February 2008, which is the Namibian tax year.

Annual net income	No.	Col. %	Cum. %	
< N\$ 50,000	59	16.1	16.1	
N\$ 50,001 to N\$ 150,000	101	27.5	43.6	
N\$ 150,001 to N\$ 250,000	91	24.8	68.4	
N\$ 250,001 to N\$ 350,000	44	12.0	80.4	
> N\$ 350,000	72	19.6	100.0	
Total	367	100.0		

Table 35: Income derived from cattle farming. Displayed is the fraction of annual net income in the period 1st March 2007 to 29th February 2008, which is the Namibian tax year.

Fraction of income from	No.	Col. %	Cum. %	
cattle farming				
0%	24	6.3	6.3	
1 to 20%	47	12.2	18.5	
21 to 40%	50	13.0	31.5	
41 to 60%	69	18.0	49.5	
61 to 80%	82	21.4	70.8	
81 to 100%	112	29.2	100.0	
Total	384	100.0		

Non-cattle on-farm	Col. %	Cum. %
income sources		
Bioenergy	1.2	1.2
Game farming	3.3	4.5
Maize	1.2	5.7
Small stock	5.9	11.6
Tourism	7.8	19.4
Other sources	2.9	22.3
Total		

Table 36: Income deriving from non-cattle on-farm income sources. Displayed is the fraction of annual net income averaged across all farmers, thus only percentages and not number of observations are given for each category. 398 farmers responded to this question. The relevant period is 1st March 2007 to 29th February 2008, which is the Namibian tax year.

Table 37: Income deriving from off-farm sources. Displayed is the fraction of annual net income in the period 1st March 2007 to 29th February 2008, which is the Namibian tax year.

Income fraction from	No.	Col. %	Cum. %	
off-farm sources				
0%	146	38.9	38.9	
1 to 20%	85	22.7	61.6	
21 to 40%	35	9.3	70.9	
41 to 60%	43	11.5	82.4	
61 to 80%	31	8.3	90.7	
81 to 100%	35	9.3	100.0	
Total	375	100.0		

Table 38: Responses from risk experiments with hypothetical rewards in the questionnaire. The experiments are detailed in Olbrich et al. (2009). Denoted are raw intervals of the coefficient of relative risk aversion (CRRA) which are calculated without considering any individual characteristics. Raw data contained an artefact where many farmers who received a mail-in questionnaire indicated extreme responses, i.e. the highest or lowest category of CRRA-intervals. In contrast, farmers who complete the questionnaire in the presence of a researcher while participating in the in-field experiments did not show this behaviour. The issue is discussed at greater length in Olbrich et al. (2011b). We here display corrected data where the highest and lowest category contain only responses of farmers who participated in the in-field experiments, scaled up to the number of respondents who did not indicate extreme responses.

Open CRRA-interval	No.	Col. %	Cum. %	
6.32, ∞	22	8.7	8.7	
4.38, 6.32	4	1.6	10.3	
2.79, 4.38	9	3.6	13.8	
1.37, 2.79	56	22.1	36.0	
0.00, 1.37	93	36.8	72.7	
-1.40, 0.00	52	20.6	93.3	
-∞, -1.40	17	6.7	100.0	
Total	253	100.0		

Table 39: Responses from in-field risk experiments with monetary reward. The experiments are detailed in Olbrich et al. (2009). Denoted are raw intervals of the coefficient relative risk aversion (CRRA) which are calculated without considering any individual characteristics.

Open CRRA-interval	No.	Col. %	Cum. %	
8.27, ∞	7	17.9	17.9	_
4.79, 8.27	0	0.0	17.9	
3.58, 4.79	0	0.0	17.9	
2.93, 3.58	0	0.0	17.9	
2.18, 2.93	1	2.6	20.5	
1.70, 2.18	0	0.0	20.5	
1.35, 1.70	10	25.6	46.2	
1.05, 1.35	2	5.1	51.3	
0.78, 1.05	6	15.4	66.7	
0.52, 0.78	2	5.1	71.8	
0.27, 0.52	1	2.6	74.4	
0.00, 0.27	6	15.4	89.7	
-0.29, 0.00	2	5.1	94.9	
-0.61, -0.29	0	0.0	94.9	
-0.99, -0.61	0	0.0	94.9	
-1.46, -0.99	0	0.0	94.9	
-∞, -1.46	2	5.1	100.0	
Total	39	100.0		

Open discount rate	No.	Col. %	Cum. %
interval			
0%, 10%	32	11.4	11.4
10%, 20%	63	22.5	33.9
20%, 30%	89	31.8	65.7
30%, 40%	48	17.1	82.9
40%, 50%	42	15.0	97.9
50%, ∞%	6	2.1	100.0
Total	280	100.0	

Table 40: Responses from time experiments with hypothetical rewards in the questionnaire. The experiments are detailed in Olbrich et al. (2009). Denoted are raw intervals of the discount rate which are calculated without considering any individual characteristics. Similar to the hypothetical risk experiments, raw data also contained an artefact. We thus present corrected data as discussed in Table 38.

Table 41: Responses from in-field time experiments with monetary rewards. The experiments are detailed in Olbrich et al. (2009). Denoted are raw intervals of the discount rate which are calculated without considering any individual characteristics.

Open discount rate	No.	Col. %	Cum. %	
interval				
0.0%, 2.5%	0	0.0	0.0	
2.5%, 5.0%	0	0.0	0.0	
5.0%, 7.5%	0	0.0	0.0	
7.5%, 10.0%	4	10.3	10.3	
10.0%, 12.5%	3	7.7	17.9	
12.5%, 15.0%	7	17.9	35.9	
15.0%, 17.5%	6	15.4	51.3	
17.5%, 20.0%	3	7.7	59.0	
20.0%, 22.5%	3	7.7	66.7	
22.5%, 25.0%	3	7.7	74.4	
25.0%, 27.5%	3	7.7	82.1	
27.5%, 30.0%	3	7.7	89.7	
30.0%, 32.5%	2	5.1	94.9	
32.5%, 35.0%	1	2.6	97.4	
35.0%, 37.5%	0	0.0	97.4	
37.5%, 40.0%	0	0.0	97.4	
40.0%, 42.5%	0	0.0	97.4	
42.5%, 45.0%	0	0.0	97.4	
45.5%, 47.5%	0	0.0	97.4	
47.5%, 50.0%	0	0.0	97.4	
50.0%, ∞%	1	2.6	100.0	
Total	39	100.0		

Time horizon	No.	Col. %	Cum. %	
Only current generation	32	8.7	8.7	
One generation	60	16.4	25.1	
Two generations	145	39.6	64.8	
Three generations	32	8.7	73.5	
Four generations	15	4.1	77.6	
Five generations	20	5.5	83.1	
Six to nine generations	3	0.8	83.9	
Ten generations or more	59	16.1	100.0	
Total	366	100.0		

Table 42: Time horizon over which grazing capacity and income should be sustained. Denoted is the number of generations beyond the own generation of the farmer.

Table 43: Other objects that should be sustained on farm besides grazing capacity and income. This was an open question in the survey, where more than one answer was possible. 286 farmers gave at least one answer. Reported are the five most frequent answers.

Objects to be sustained	No.	Col. %
Groundwater level	105	36.7
Ecosystem quality	79	27.6
Social conditions	44	15.4
Political conditions	41	14.3
Infrastructure	30	10.5

Appendix D: Cluster analysis



Figure 5: Dendrogram for three cluster solution. Cluster labels and observations per cluster are indicated above the respective branch. Clusters are MULTOWN (multiple owners), ENFIMA (favorable environment / low financial risk management) and AFRIKAANS (Afrikaans farmers). N=108.

Table 44: Results for pseudo F- and pseudo T square-indices for different numbers of clusters. Good numb	er of
clusters are indicated by high values for pseudo F-index and by low values for pseudo T-square index.	

Number of	pseudo F	pseudo T
clusters		square
1		0.30
2	0.30	0.98
3	0.57	0.00
4	0.38	0.99
5	0.53	1.26
6	0.59	1.48
7	0.79	3.07
8	1.22	2.43
9	1.59	0.15
10	1.41	1.44
11	1.43	0.35
12	1.31	0.07
13	1.20	0.02
14	1.09	1.56
15	1.03	0.17

Table 45: Cluster-averaged values of characteristics for clusters MULTOWN (multiple owners), ENFIMA (favorable environment / low financial risk management) and AFRIKAANS (Afrikaans farmers). p-values for cluster differences calculated for each characteristic by one-way ANOVA for continuous and Chi-square test for binary characteristics. Shading indicates cluster responsible for differences as calculated by Bonferroni-corrected t-tests for continuous and pair-wise Chi-square test for binary characteristics, with the significance levels: *** p<0.01, ** p<0.05, * p<0.1. Dark shading denotes that cluster differs from both other clusters, light grey shading that it differs from only one other cluster (the one most different in averaged values). N=108.

Clusters:	MULTOWN	EnFIMA	IMA AFRIKAANS p-value	
1) Personal and farm features				
Household size [number of members]	3.6	2.7^{*}	3.6	0.036
Age [years]	46.9***	55.5	51.4	0.010
Afrikaans [%]	19.4	7.7	95.7***	0.000
Education level [1=no high school graduation, 6=Doctorate]	3.8	4.0	3.4**	0.035
Weekend farmer [%]	83.3	80.8	87.0	0.773
Average rainy season assessment [1=very poor, 6=very good]	3.9	4.1	3.9	0.328
Single owners [%]	41.7***	84.6	89.1	0.000
Rangeland [hectare]	9,448	7,980	8,181	0.483
Land net rented [hectare]	2,587**	512	919	0.010
Land quality[1=very poor quality, 6=very good quality]	4.0	4.2	4.4	0.342
Actual bush cover [1=0%, 6=81 to 100%]	3.7	3.3	3.6	0.392
Optimal bush cover [%]	23.8	19.3	26.5	0.115
Grazing capacity [Large Stock Unit per hectare]	0.071	0.089^{*}	0.074	0.016
Oxen production [%]	68.3	60.5	42.3***	0.008
Annual net income [1 = <n\$50,000, 6="">N\$350,000]</n\$50,000,>	2.9	3.4	3.2	0.358
2) Risks faced by farmers [1=no risk, 6=very high risk]				
Rainfall	4.9	4.4*	5.0	0.067
3) Risk management strategies [1=not at all important, 6=very important]	_			
On-farm management strategies				
Purchase of supplementary feed	4.3	4.8	4.5	0.371
Choice of cattle production system	4.9	3.9**	4.5	0.039
Choice of breed adapted to high variability in grass production	4.8	3.9**	4.6	0.032
Resting part of rangeland in good rainy seasons	4.1**	4.7	5.0	0.025
Purchase/lease of rangeland for scale effects	4.1	2.7^{*}	3.3	0.004
Financial management strategies				
Advances on livestock sales	3.6**	1.4**	2.6**	0.000
Checking account as financial buffer	4.8	4.3	4.8	0.327
Loans for covering operating losses	3.0	1.5**	* 3.4	0.000
Income from off-farm employment or assets	3.9	3.6	3.8	0.803
Investment into agricultural derivatives	2.4	1.9	2.0	0.392

Characteristics (continued)	MULTOWN	EnFiMa	AFRIKAANS	p-value
4) Individual risk and time preferences				
Risk preference index [1=very risk averse, 7=very risk attracted]	4.6	5.0	4.7	0.416
Time preference index [1=very patient, 7=very impatient]	3.1	2.6*	3.2	0.069
5) Normative views of sustainability			-	
Sustainable annual net income [N\$]	292,806	251,539	294,000	0.567
Sustainable grazing capacity [Large Stock Unit per hectare]	0.074	0.086	0.076	0.217
Acceptable income risk [probability]	0.6	0.5	0.6	0.801
Acceptable grazing capacity risk [probability]	0.6	0.7^{**}	0.6	0.009
Time horizon [generations]	3.3	4.1	3.5	0.671
Selected characteristics not included in cluster analysis				
Female [%]	2.8	3.9	4.3	
Farm experience [years]	17.8	22.7	23.7	
Regional location of farm [%]				
Khomas	19.4	23.1	17.4	
Omaheke	8.3	26.9	23.9	
Otjozondjupa	58.3	38.4	30.4	
Total on-farm precipitation rainy season 2006/07 [mm]	241.6	291.4	242.2	
Total on-farm precipitation rainy season 2007/08 [mm]	439.2	441.0	395.7	
Land owned [ha]	7,651	7,352	7,696	
Land operated [ha]	10,187	8,028	8,600	
Own cattle, Apr 2008	521	492	454	
Income from cattle farming $[1=0\%, 6=81-100\%]$	4.3	4.6	4.4	
NAU member [%]	91.7	92.3	82.6	
Emerging commercial farmer [%]	16.6	8.3	2.2	
Risk management strategies [1=not at all important, 6=very important]				
Purchase/lease of rangeland for spatial diversification	3.7	2.4	3.0	
Forwards contracts	4.1	2.7	3.3	

References

- Andersen, S., Harrison, G.W., Lau, M.I., Rutström, E.E., 2006. Elicitation using multiple price list formats. Experimental Economics 9, 383–405.
- Backhaus, K., Erichson, B., Plinke, W., Weiber, R., 2006. Multivariate Analysemethoden: Eine anwendungsorientierte Einführung, 11th ed. Springer, Berlin, Germany.
- Baumgärtner, S., Quaas, M.F., 2009. Ecological-economic viability as a criterion of strong sustainability under uncertainty. Ecological Economics 68, 2008–2020.
- Burke, A., 1997. The impact of large herbivores on floral composition and vegetation structure in the Naukluft Mountains, Namibia. Biodiversity and Conservation 6, 1203–1217.
- Calinski, T., Harabasz, J., 1974. A dendrite method for cluster analysis. Communications in Statistics 3, 1–27.
- CBS Namibia, [Central Bureau of Statistics Namibia], 2006. Namibia Household Income and Expenditure Survey 2003/2004, Preliminary Report. Central Bureau of Statistics, Windhoek, Namibia.
- CBS Namibia, [Central Bureau of Statistics Namibia], 2010. Namibia Consumer Price Index – October 2010. Central Bureau of Statistics, Windhoek, Namibia.
- Chapman, G., 2010. Popular perception and climate change: Mapping the varying experience of precipitation. Centre for Advanced Study, Oslo, Norway. (Internet publication, http://www.cas.uio.no/publications_/transference.php).
- CIA, [Central Intelligence Agency], 2011. The World Factbook 2011. https://www.cia.gov/library/publications/the-world-factbook/. Accessed 09th June 2011.
- de Klerk, J., 2004. Bush Encroachment in Namibia. Report on Phase 1 of the Bush Encroachment Research, Monitoring and Management Project. Ministry of Environment and Tourism, Windhoek, Namibia.
- Duda, R.O., Hart, P.E., 1973. Pattern classification and scene analysis. John Wiley and Sons, New York, United States.
- EIA, [Energy Information Administration], 2011. Database of spot prices for WTI crude oil, Cushing, Oklahoma. http://www.eia.gov/. Accessed 07th October 2011.
- Fynn, R., O'Connor, T., 2000. Effect of stocking rate and rainfall on rangeland dynamics and cattle performance in a semi-arid savanna, South Africa. Journal of Applied Ecology 37, 491–507.

- Gower, J., 1971. A general coefficient of similarity and some of its properties. Biometrics 27, 857–871.
- Harrison, G.W., Humphrey, S.J., Verschoor, A., 2010. Choice under uncertainty: Evidence from Ethiopia, India and Uganda. Economic Journal 120, 80–104.
- Harrison, G.W., Lau, M.I., Rutström, E.E., Sullivan, M., 2005. Eliciting risk and time preferences using field experiments: Some methodological issues, in: Carpenter, J.P., Harrison, G.W., List, J.A. (Eds.), Field experiments in economics, 1st ed. Elsevier, Amsterdam, Netherlands, pp. 125–218.
- Howarth, R.B., 2007. Towards an operational sustainability criterion. Ecological Economics 63, 656–663.
- Isaacs, D., 2007. Tenth arrest for Khorixas stock theft. The Namibian, Windhoek, Namibia, 15th January 2007.
- MAWF, [Ministry of Agriculture, Water and Forestry], 2009. Agricultural statistics bulletin (2000–2007). Ministry of Agriculture, Water and Forestry, Windhoek, Namibia.
- MEA, [Millenium Ecosystem Assessment], 2005. Chapter 22: Dryland systems, in: Hassan, R., Scholes, R., Ash, N. (Eds.), Ecosystem and human well-being. Current state and trends, Volume 1. Island Press, Washington, D.C., United States.
- Meat Board, [Meat Board of Namibia], unpublished. Monthly cattle prices 2008. Meat Board of Namibia, Windhoek, Namibia.
- Mendelsohn, J., 2006. Farming systems in Namibia. Research and Information Services in Namibia, Windhoek, Namibia.
- Meyn, M., 2007. The end of the current EU preferences for Namibia: Economic and social impacts. Overseas Development Institute, London, United Kingdom.
- NMS, [Namibia Meteorological Service], unpublished. Daily precipitation data for 160 weather stations for the period 1913–2008. Namibian Meteorological Service, Windhoek, Namibia.
- Olbrich, R., 2012. Environmental risk and sustainability: The case of commercial livestock farming in semi-arid rangelands. Dissertation, Leuphana University Lüneburg, Germany.
- Olbrich, R., Quaas, M.F., Baumgärtner, S., 2009. Sustainable use of ecosystem services under multiple risks – A survey of commercial cattle farmers in semi-arid rangelands in Namibia. Working Paper Series in Economics No. 137, Leuphana University Lüneburg, Germany.

- Olbrich, R., Quaas, M.F., Baumgärtner, S., 2011a. Personal norms of sustainability and their impact on management The case of rangeland management in semi-arid regions.
 Working Paper Series in Economics No. 209, Leuphana University Lüneburg, Germany.
- Olbrich, R., Quaas, M.F., Baumgärtner, S., in press. A survey of commercial cattle farmers in semi-arid rangelands of Namibia on risk, sustainability and management. Schmollers Jahrbuch / Journal of Applied Social Science Studies.
- Olbrich, R., Quaas, M.F., Haensler, A., Baumgärtner, S., 2011b. Risk preferences under heterogeneous environmental risk. Working Paper Series in Economics No. 208, Leuphana University Lüneburg, Germany.
- Ostrom, E., 1999. Revisiting the commons: Local lessons, global challenges. Science 284, 278–282.
- Quaas, M.F., Baumgärtner, S., 2008. Natural vs. financial insurance in the management of public-good ecosystems. Ecological Economics 65, 397–406.
- Schutz, W., 2010. Personal communication, interview conducted on 01st March 2010, Meat Board of Namibia, Windhoek, Namibia.
- Schwartz, S.H., 1973. Normative explanations of helping behavior: A critique, proposal and empirical test. Journal of Experimental Social Psychology 9, 349–364.
- Schwartz, S.H., 1977. Normative influences on altruism. Advances in Experimental Social Psychology 10, 221–279.
- Shogren, J.F., Crocker, T.D., 1999. Risk and its consequences. Journal of Environmental Economics and Management 37, 44–51.
- Sweet, J., 1998. Livestock Coping with Drought: Namibia A case study. Pasture Development Network, Overseas Development Institute, London, United Kingdom.
- Unganai, L., 1996. Historic and future climatic change in Zimbabwe. Climate Research 6, 137–145.
- Ward, J.H., 1963. Hierarchical grouping to optimize an objective function. Journal of the American Statistical Association 58, 236–244.
- Wiegand, K., 2010. In the rhythm of the vegetation. German Research 32, 12–16.

Working Paper Series in Economics

(recent issues)

No.247:	Alexander Vogel and Joachim Wagner: Exports, R&D and Productivity in German Business Services Firms: A test of the Bustos-model, August 2012
No.246:	<i>Alexander Vogel</i> and <i>Joachim Wagner</i> : Innovations and Exports of German Business Services Enterprises: First evidence from a new type of firm data, August 2012
No.245:	Stephan Humpert: Somewhere over the Rainbow: Sexual Orientation Discrimination in Germany, July 2012
No.244:	<i>Joachim Wagner</i> : Exports, R&D and Productivity: A test of the Bustos-model with German enterprise data, June 2012 [published in: Economics Bulletin, 32 (2012), 3, 1942-1948]
No.243:	<i>Joachim Wagner</i> . Trading many goods with many countries: Exporters and importers from German manufacturing industries, June 2012
No.242:	<i>Joachim Wagner</i> : German multiple-product, multiple-destination exporters: Bernard- Redding-Schott under test, June 2012 [published in: Economics Bulletin, 32 (2012), 2, 1708-1714]
No.241:	Joachim Fünfgelt and Stefan Baumgärtner: Regulation of morally responsible agents with motivation crowding, June 2012
No.240:	John P. Weche Gelübcke: Foreign and Domestic Takeovers: Cherry-picking and Lemon-grabbing, April 2012
No.239:	Markus Leibrecht and Aleksandra Riedl: Modelling FDI based on a spatially augmented gravity model: Evidence for Central and Eastern European Countries, April 2012
No.238:	Norbert Olah, Thomas Huth und Dirk Löhr. Monetarismus mit Liquiditätsprämie Von Friedmans optimaler Inflationsrate zur optimalen Liquidität, April 2012
No.237:	<i>Markus Leibrecht</i> and <i>Johann Scharler</i> : Government Size and Business Cycle Volatility; How Important Are Credit Contraints?, April 2012
No.236:	<i>Frank Schmielewski</i> and <i>Thomas Wein</i> : Are private banks the better banks? An insight into the principal-agent structure and risk-taking behavior of German banks, April 2012
No.235:	Stephan Humpert: Age and Gender Differences in Job Opportunities, March 2012
No.234:	Joachim Fünfgelt and Stefan Baumgärtner. A utilitarian notion of responsibility for sustainability, March 2012
No.233:	<i>Joachim Wagner</i> . The Microstructure of the Great Export Collapse in German Manufacturing Industries, 2008/2009, February 2012
No.232:	<i>Christian Pfeifer</i> and <i>Joachim Wagner</i> . Age and gender composition of the workforce, productivity and profits: Evidence from a new type of data for German enterprises, February 2012
No.231:	Daniel Fackler, Claus Schnabel, and Joachim Wagner. Establishment exits in Germany: the role of size and age, February 2012
No.230:	Institut für Volkswirtschaftslehre: Forschungsbericht 2011, January 2012
No.229:	<i>Frank Schmielewski</i> : Leveraging and risk taking within the German banking system: Evidence from the financial crisis in 2007 and 2008, January 2012
No.228:	<i>Daniel Schmidt</i> and <i>Frank Schmielewski</i> : Consumer reaction on tumbling funds – Evidence from retail fund outflows during the financial crisis 2007/2008, January 2012

- No.227: Joachim Wagner. New Methods for the Analysis of Links between International Firm Activities and Firm Performance: A Practitioner's Guide, January 2012
- No.226: Alexander Vogel and Joachim Wagner: The Quality of the KombiFiD-Sample of Business Services Enterprises: Evidence from a Replication Study, January 2012
- No.225: Stefanie Glotzbach: Environmental justice in agricultural systems. An evaluation of success factors and barriers by the example of the Philippine farmer network MASIPAG, January 2012
- No.224: Joachim Wagner: Average wage, qualification of the workforce and export performance in German enterprises: Evidence from KombiFiD data, January 2012 [published in: Journal for Labour Market Research, 45 (2012), 2, 161-170]
- No.223: *Maria Olivares* and *Heike Wetzel*: Competing in the Higher Education Market: Empirical Evidence for Economies of Scale and Scope in German Higher Education Institutions, December 2011
- No.222: Maximilian Benner: How export-led growth can lead to take-off, December 2011
- No.221: Joachim Wagner and John P. Weche Gelübcke: Foreign Ownership and Firm Survival: First evidence for enterprises in Germany, December 2011
- No.220: *Martin F. Quaas, Daan van Soest,* and *Stefan Baumgärtner*. Complementarity, impatience, and the resilience of natural-resource-dependent economies, November 2011
- No.219: *Joachim Wagner*: The German Manufacturing Sector is a Granular Economy, November 2011 [published in: Applied Economics Letters, 19(2012), 17, 1663-1665]
- No.218: Stefan Baumgärtner, Stefanie Glotzbach, Nikolai Hoberg, Martin F. Quaas, and Klara Stumpf: Trade-offs between justices, economics, and efficiency, November 2011
- No.217: *Joachim Wagner*: The Quality of the KombiFiD-Sample of Enterprises from Manufacturing Industries: Evidence from a Replication Study, November 2011
- No.216: John P. Weche Gelübcke: The Performance of Foreign Affiliates in German Manufacturing: Evidence from a new Database, November 2011
- No.215: Joachim Wagner. Exports, Foreign Direct Investments and Productivity: Are services firms different?, September 2011
- No.214: Stephan Humpert and Christian Pfeifer. Explaining Age and Gender Differences in Employment Rates: A Labor Supply Side Perspective, August 2011
- No.213: John P. Weche Gelübcke: Foreign Ownership and Firm Performance in German Services: First Evidence based on Official Statistics, August 2011 [forthcoming in: The Service Industries Journal]
- No.212: John P. Weche Gelübcke: Ownership Patterns and Enterprise Groups in German Structural Business Statistics, August 2011 [published in: Schmollers Jahrbuch / Journal of Applied Social Science Studies, 131(2011), 4, 635-647]
- No.211: *Joachim Wagner*: Exports, Imports and Firm Survival: First Evidence for manufacturing enterprises in Germany, August 2011
- No.210: Joachim Wagner: International Trade and Firm Performance: A Survey of Empirical Studies since 2006, August 2011 [published in: Review of World Economics, 2012, 148 (2), 235-267]
- No.209: Roland Olbrich, Martin F. Quaas, and Stefan Baumgärtner. Personal norms of sustainability and their impact on management – The case of rangeland management in semi-arid regions, August 2011

- No.208: Roland Olbrich, Martin F. Quaas, Andreas Haensler and Stefan Baumgärtner: Risk preferences under heterogeneous environmental risk, August 2011
- No.207: Alexander Vogel and Joachim Wagner. Robust estimates of exporter productivity premia in German business services enterprises, July 2011 [published in: Economic and Business Review, 13 (2011), 1-2, 7-26]
- No.206: Joachim Wagner: Exports, imports and profitability: First evidence for manufacturing enterprises, June 2011
- No.205: Sebastian Strunz: Is conceptual vagueness an asset? Resilience research from the perspective of philosophy of science, May 2011
- No.204: Stefanie Glotzbach: On the notion of ecological justice, May 2011
- No.203: Christian Pfeifer: The Heterogeneous Economic Consequences of Works Council Relations, April 2011
- No.202: Christian Pfeifer, Simon Janssen, Philip Yang and Uschi Backes-Gellner. Effects of Training on Employee Suggestions and Promotions in an Internal Labor Market, April 2011
- No.201: Christian Pfeifer: Physical Attractiveness, Employment, and Earnings, April 2011
- No.200: Alexander Vogel: Enthüllungsrisiko beim Remote Access: Die Schwerpunkteigenschaft der Regressionsgerade, März 2011
- No.199: *Thomas Wein*: Microeconomic Consequences of Exemptions from Value Added Taxation The Case of Deutsche Post, February 2011
- No.198: *Nikolai Hoberg* and *Stefan Baumgärtner*. Irreversibility, ignorance, and the intergenerational equity-efficiency trade-off, February 2011
- No.197: Sebastian Schuetz: Determinants of Structured Finance Issuance A Cross-Country Comparison, February 2011
- No.196: Joachim Fünfgelt and Günther G. Schulze: Endogenous Environmental Policy when Pollution is Transboundary, February 2011
- No.195: *Toufic M. El Masri*: Subadditivity and Contestability in the Postal Sector: Theory and Evidence, February 2011
- No.194: *Joachim Wagner*: Productivity and International Firm Activities: What do we know?, January 2011 [published in: Nordic Economic Policy Review, 2011 (2), 137-161]
- No.193: *Martin F. Quaas* and *Stefan Baumgärtner*. Optimal grazing management rules in semiarid rangelands with uncertain rainfall, January 2011
- No.192: Institut für Volkswirtschaftslehre: Forschungsbericht 2010, Januar 2011
- No.191: Natalia Lukomska, Martin F. Quaas and Stefan Baumgärtner. Bush encroachment control and risk management in semi-arid rangelands, December 2010
- No.190: *Nils Braakmann:* The causal relationship between education, health and health related behaviour: Evidence from a natural experiment in England, November 2010
- No.189: *Dirk Oberschachtsiek and Britta Ulrich:* The link between career risk aversion and unemployment duration: Evidence of non-linear and time-depending pattern, October 2010

(see www.leuphana.de/institute/ivwl/publikationen/working-papers.html for a complete list)

Leuphana Universität Lüneburg Institut für Volkswirtschaftslehre Postfach 2440 D-21314 Lüneburg Tel.: ++49 4131 677 2321 email: brodt@leuphana.de

www.leuphana.de/institute/ivwl/publikationen/working-papers.html